Author's response to reviews

Title: Expression of CXCR4 and Breast Cancer Prognosis A Systematic Review and Meta-analysis

Authors:

Zhigang Zhang (zzg2011@zju.edu.cn)
Chao Ni (zzg201310@sina.com)
Wuzhen Chen (474355615@qq.com)
Ping Wu (zzg201110@163.com)
Zhen Wang (zzg201310@gmail.com)
Junhua Yin (zzg201310@sina.com)
Jian Huang (zzg200610@126.com)
Fuming Qiu (zzg200610@sina.com)

Version: 2
Date: 23 November 2013

Author's response to reviews:

1. For Referee 1:
The reviewer’s report has no questions.

2. For Referee 2:
The first question, we did not consider the population in USA which ethnicity is heterogeneous, because the original research didn’t provide this heterogeneous. In our paper we just carried the distinction of different regions.
The second question, we have revised the serial number of references in Table 1.